Author's response to reviews

Title: Cancer stem-like sphere cells induced from de-differentiated hepatocellular carcinoma-derived cell lines possess the resistance to anti-cancer drugs

Authors:

Noriaki Hashimoto (nh1219@yamaguchi-u.ac.jp)
Ryouichi Tsunedomi (tsune-r@yamaguchi-u.ac.jp)
Kiyoshi Yoshimura (kyoshim1@yamaguchi-u.ac.jp)
Yusaku Watanabe (yusakuw@yamaguchi-u.ac.jp)
Masaaki Oka (2geka-1@po.cc.yamaguchi-u.ac.jp)

Version: 2
Date: 9 April 2014

Author's response to reviews: see over
Dear Journal Editorial Office

(\textit{BMC Cancer})

Thank you so much for your kind invitation to submit our revised manuscript.

We here submit the revised manuscript, together with point-by-point account of how we have responded. We sincerely hope that this manuscript meets with your approval.

1. Title Page: Please include email addresses of all authors.
A1. We added email addresses of all authors in title page.

2. Line Numbering.
A2. Revised manuscript include line and page numbers without insertion of page breaks.

Thank you very much for your consideration.

Sincerely yours

Ryoui\-chi Tsunedomi, Ph.D.
Assistant Professor
Department of Digestive Surgery and Surgical Oncology,
Yamaguchi University Graduate School of Medicine,
1-1-1 Minami-Kogushi, Ube, Yamaguchi 755-8505, JAPAN
Tel: +81-836-22-2264
Fax: +81-836-22-2263
e-mail: tsune-r@yamaguchi-u.ac.jp